

Kent Academic Repository

Idris, Bochra, Saridakis, George and Khan, Zaheer (2022) The effect of outward and inward internationalisation on different types of innovation: Evidence from UK SMEs. Journal of International Management, 28 (2). ISSN 1075-4253.

Downloaded from

https://kar.kent.ac.uk/91772/ The University of Kent's Academic Repository KAR

The version of record is available from

https://doi.org/10.1016/j.intman.2021.100903

This document version

Author's Accepted Manuscript

DOI for this version

Licence for this version

CC BY-NC-ND (Attribution-NonCommercial-NoDerivatives)

Additional information

Versions of research works

Versions of Record

If this version is the version of record, it is the same as the published version available on the publisher's web site. Cite as the published version.

Author Accepted Manuscripts

If this document is identified as the Author Accepted Manuscript it is the version after peer review but before type setting, copy editing or publisher branding. Cite as Surname, Initial. (Year) 'Title of article'. To be published in *Title* of *Journal*, Volume and issue numbers [peer-reviewed accepted version]. Available at: DOI or URL (Accessed: date).

Enquiries

If you have questions about this document contact ResearchSupport@kent.ac.uk. Please include the URL of the record in KAR. If you believe that your, or a third party's rights have been compromised through this document please see our Take Down policy (available from https://www.kent.ac.uk/guides/kar-the-kent-academic-repository#policies).

The effect of outward and inward internationalisation on different types of innovation: evidence from UK SMEs

Bochra Idris*

Aston University, College of Business and Social Sciences, Birmingham, B4 7ET, UK b.idris@aston.ac.uk

George Saridakis

The University of Kent, Kent Business School, Canterbury, CT2 7NZ, UK G.Saridakis@kent.ac.uk

Zaheer Khan

University of Aberdeen Business School, King's College, Aberdeen, AB24 3FX, UK
School of Marketing and Communication, University of Vaasa, Finland
zaheer.khan@abdn.ac.uk

*Corresponding Author

The effect of outward and inward internationalisation on different types of innovation: evidence from UK SMEs

Abstract

This paper empirically examines the effect of outward internationalisation, inward internationalisation activities and outward-inward internationalisation (i.e., exporting and importing) on different types of innovation undertaken by UK SMEs. Specifically, we differentiate between product innovation and process innovation and examine the potential effect that they can generate individually and in combination. The results show that both inward and outward internationalisation support product and process innovation in SMEs. However, such an effect is found to be stronger for the combined outward-inward internationalisation operations than for the single mode undertaken by SMEs. The results are found to be robust across the different types of innovation. However, sub-sample analysis shows that, although innovation responds to different internationalisation operations in micro and small firms, for medium-sized firms, only the combination of outward and inward internationalisation operations increases the probability of undertaking both innovations. Using the organisational learning theory, we argue that engaging in both internationalisation activities simultaneously enables firms to acquire a more diverse and richer set of knowledge and key information - through double loop learning, which is translated into increased levels of innovation. Hence, our results have important theoretical, managerial, and policy implications and stimulate the existing debate in the area.

Keywords: SMEs, outward internationalisation, inward internationalisation, learning, innovation, product innovation, process innovation.

1. Introduction

There has been increasing interest in understanding the role of internationalisation on firms' innovation. Existing research shows that both internationalisation and innovation contribute to firms' performance, productivity and growth (Prashantham, 2008; Halilem et al., 2014; Abubakar et al., 2019). Since internationalisation and innovation are interlinked, increasing attention is being directed to the specific channels through which international trade can affect firm innovation (Altomonte et al., 2014). To this end, the international business (IB) literature suggests that firms that are engaged in international activities are more likely to gain substantial competitive advantages through the interaction with their customers and suppliers and the competitive conditions of the foreign market in which they operate (Salomon & Shaver, 2005; Damijan & Kostevc, 2010). In addition, the organisational learning (OL) theory suggests that firms that are exposed to different economic environments and conditions can enhance their creativity and innovative activities to overcome the competition and strengthen their market share through the acquisition and leveraging of diverse sources of knowledge (cf. March, 1991; Argote, 2011; Puthusserry et al., 2020). It can be argued, therefore, that internationalisation in the form of imports and/or exports can serve as a learning opportunity for firms to gain more market and organisational knowledge, which in turn can trigger innovation (Kiriyama, 2012). Especially for smaller firms, since they face limitations in their internal resources as well as suffer from liabilities of smallness and newness compared with larger firms (Stinchcombe, 1965; Aldrich & Auster 1986; Gimenez-Fernandez et al., 2020). Thus, innovative activities may be affected more by knowledge factors that are external to the firm (Acs, 2002; Abubakar & Mitra, 2009; Abubakar et al., 2019).

Therefore, exposure to international markets can allow firms to develop and expand their set of capabilities and thus increase their likelihood of growth and enhance their dominance in the marketplace (Lu & Beamish, 2006; Prashantham, 2008; Zahra et al., 2018). For example, internationalised firms may be more likely to be exposed to new organisational ideas and methods of production, providing them with greater learning opportunities and knowledge tools to develop new skills and expand or improve existing organisational approaches that are not available in their domestic market (Hitt et al., 1997; Jones, 2001; Puthusserry et al., 2020). Although a great number of studies provide evidence regarding the positive effect of internationalisation on innovation (e.g., Kafouros et al., 2008; Lecerf, 2012), most of these studies concentrate on larger firms (e.g., Kafouros et al., 2008) rather than on SMEs (Abubakar et al., 2019). This is perhaps surprising, since previous research shows that SMEs that are involved in international activities are 'three times' more likely to introduce

products or services that are new to their sector than those SMEs that focus entirely on their domestic market (European Commission, 2010; Love & Roper, 2015). Although the available literature stresses the link between internationalisation and innovation, it is still in its infant stage regarding the individual and combined roles of outward and inward internationalisation activities on innovation.

Specifically, most of the previous studies focus either on the link between outward internationalisation and innovation (e.g., Salomon & Shaver, 2005; Love & Ganotakis, 2013; Golovko & Valentini, 2014; Olabisi, 2017; Fassio, 2018) or on the effect of inward internationalisation on innovation (e.g., Goldberg et al., 2010; Grosse & Fonseca, 2012; Kiriyama, 2012; Liu & Qiu, 2016; Chen et al., 2017) to determine how firms accumulate and implement knowledge (Hernández & Nieto, 2016). However, this literature ignores the potential effect on innovation that can be generated by combining the two forms of internationalisation activities. To address this gap, this paper investigates whether a combination of outward and inward internationalisation operations along with their individual effects can help SMEs to develop product and process innovations and, if so, to what extent. Theoretically, we draw insights from the OL theory (Senge, 1990; Argote, 2011; Gerschewski et al., 2018), and apply it to propose that the knowledge flow from international activities can stimulate SMEs' innovation (Zahra et al., 2009). In contrast to a large body of previous work, we do not treat innovation as an overall construct capturing different types of innovation together (Azari et al., 2017). In addition, instead of focusing on different types of innovation, separately, (e.g., Alegre et al., 2012; Bratti & Felice, 2012; D'Angelo et al., 2013; Boermans & Roelfsema, 2016; Martínez-Román et al., 2019, for product innovation; and Damijan & Kostevc, 2010, for process innovation; and Cassiman & Golovko, 2011; Monreal-Pérez et al., 2012, Damijan & Kostevc, 2015; Abubakar et al., 2019; Henly & Song, 2020; for product and process innovation, separately), we follow limited but growing recent research (e.g., Hullova et al., 2016; Saridakis et al., 2019) and consider a combination of different types of innovation (i.e., a combination of product and process) and their potential association with inward and outward internationalisation activities.

Such an approach provides a comprehensive and holistic understanding about the impact of inward-outward internationalisation activities and their effect on different types of innovation. The international process view and internationalisation studies primarily focus has been on firms' outward activities in the development of market knowledge (cf. Johanson & Vahlne, 1977, 1990), whereas scholars have also suggested that other sources of knowledge can be important for value creations (e.g., Jones, 2001; Forsgren, 2002; Puthusserry et al.,

2020). In such a context examining the connections between outward-inward internationalisation activities and their potential impact on firm-level outcomes such as innovation has significant implications for developing a much fine-grained understanding of firms' internationalisation process. Therefore, the examination of both inward and outward internationalisation activities provide novel and systematic understanding about the coordination, generation and exploitation of different sources of knowledge by firms for different types of innovation.

In order to address the above gaps, we leverage data from the UK Small Business Survey (BEIS, 2018), conducted between August 2016 and January 2017 by BMG Research Ltd. The survey is a large-scale telephone survey conducted with 9,248 small business owner-managers (those with up to 249 employees) in the United Kingdom. By utilising such a large-scale data set, the aim is to provide a complete picture of the effects of different internationalisation operations on innovation for SMEs (e.g., Seker, 2009; Hernández & Nieto, 2016) by focusing on the SMEs originating from the UK. Moreover, the use of a large-scale data set allows us to examine the differences between larger-sized SMEs and smaller ones, which can be hidden when data is aggregated into a single size category (Idris & Saridakis, 2018). Such a fine-grained systematic analysis is rare in the extant internationalisation research (cf. Bagheri et al., 2019).

Importantly, this work contributes to three distinctive streams of literature - the international business (IB) literature (e.g., Damijan & Kostevc, 2010; Nieto & Rodríguez, 2011; Grosse & Fonseca, 2012; Hernández & Nieto, 2016; Abubakar et al., 2019; Bagheri et al., 2019), the small business (SB) literature (e.g., Andersson & Lööf, 2009, 2012; Hernández & Nieto, 2016), the OL and international knowledge literatures (e.g., Argote, 2011; Chiva et al., 2014; Gerschewski et al., 2018) - in three important ways.

First, we add to the previous IB literature by examining the effect of outward-inward internationalisation operations simultaneously on SMEs' different types of innovation. In other words, our paper extends the extant studies that highlight the importance of taking into account the complementarities that may arise from different international activities (Bertrand, 2011; Hernández & Nieto, 2016). However, existing studies have predominately focused on outward internationalisation and ignored the role of inward activities on the development of innovative capabilities that are vital for developing competitive advantages in foreign markets (e.g., Cavusgil, 1998; Eriksson et al., 2000; Johanson & Vahlne, 1977, 1990; Bagheri et al., 2019). In this way, the paper extends the typical examination of the effect of either outward (e.g., Lileeva & Trefler, 2010; Ganotakis & Love, 2011; Bratti & Felice, 2012) or inward (e.g.,

Anderson & Lööf, 2009; Filippetti et al., 2012; Shahabadi & Havaj; 2012) international operations on innovation. Studies such as ours, which integrate both inward and outward internationalisation and simultaneously examine their effect on product and process innovations, are relatively rare (cf. Bagheri et al., 2019). Second, we contribute to the literature on international knowledge spillover, which suggests that the flow of knowledge is facilitated by participating in international trade activities. However, we extend the typical examination that either exporting or importing can increase firm's accumulation of knowledge. Our results clearly suggest that undertaking both types of internationalisation activities, simultaneously, may allow firms to increase their knowledge and access to information, thereby enhancing their absorptive capacity (Yao et al., 2013), which is ultimately related to firms' performance (George et al., 2001). In this context, our study provides novel insights about the different sources of knowledge - again highlighting that both outward and inward internationalisation activities can enable SMEs to acquire and internalise knowledge, which in turn facilitate their absorptive capacity (Hernández & Nieto, 2016), and support their international operations. Beside SMEs, large firms from emerging markets are also expanding abroad by learning from inward activities and developing their capabilities that are conducive for expansion into foreign markets (cf. Li et al., 2017). Third, we contribute to the OL theory by providing support to the view that it is important for firms to acquire and share knowledge with international firms due to the demands of globalisation (Levitt & March, 1988; Argote & Miron-Spektor, 2011; Hernández & Nieto, 2016). By exploring and exploiting different sources of knowledge, small resource-constrained firms can generate both product and process innovation through double loop learning (Argyris & Schon, 1978; Huber, 1991; March, 1991; Fletcher, 2009; Argote, 2011). Prior international process studies focus on incremental learning approaches where firms can develop competitive advantage by engaging in outward international activities through the accumulation of market knowledge (Johanson & Valhne, 1977, 1990). The findings of this study contribute to the process based view of internationalisation by bringing in multidimensional - holistic view of learning through inward-outward activities and their impact on different types of innovation. Small resource constrained firms can experiment and develop experiential learning through various sources while expanding into different markets through outward activities (cf. Zahra et al. 2018; Puthusserry et al., 2020), and can also enhance their learning through networking and developing connections with international sources of knowledge (through suppliers) by pursuing inward internationalisation activities. The connections between outward-inward internationalisation and the development of innovative capabilities by internationalising SMEs can be vital for developing a complete picture of firms'

internationalisation process, which so far has received limited scholarly attention (cf. Cavusgil, 1998; Bagheri et al., 2019; Eriksson et al., 2000; Westney, 2020). By examining both activities and their connections with innovation suggest that internationalising SMEs could gain more by engaging in both activities and this broadly contributes to the wider IB literature that has focused on a single activity such as outward internationalisation at a given time (cf. Adomako et al., 2017; Bagheri et al., 2019; Puthusserry et al., 2020). Thus, these findings provide important insights and respond to recent calls to examine how inward internationalisation activities affects domestic firms (cf. Westney, 2020, p. 1197). In addition, we provide important and fine-grained understanding to the innovation literature by connecting both activities and different types of innovation in the context of SMEs, as previous literature focuses on the causeeffect type of relationships -exploring either the impact of innovation on internationalisation or vice versa (cf. Chiva et al., 2014). Lastly, we provide a much fine-grained view of inwardoutward activities across different types of firms and the impact of dual activities on different types of firms' innovation. Since firms varies in their resource base, therefore, examining inward-outward activities and their impact on different types of innovation across different types of firms provide a much needed understanding about the type of firms that may benefit from both vs. single activities. Developing capabilities and their role in internationalisation is at the centre of understanding the foreign expansion of firms as well as their survival in foreign markets, thus this study offers unique insights on this important topic.

The rest of paper is organised as follows. Section 2 reviews the existing literature and derives the set of the hypotheses to be examined in this paper. Section 3 presents the data and the measurements used in the model. Section 4 contains the statistical methods and the results of the paper. Section 5 discusses the results, and the last section concludes the paper and offers directions for future research.

2. Background and hypotheses development

2.1 Outward internationalisation and innovation

Although different modes of internationalisation are available to SMEs, exporting is still often considered a firm's initial stage of internationalisation (Jones, 2001; Golovko & Valentini, 2011). Following recent empirical studies in this area (e.g., Idris & Saridakis, 2018; Abubakar et al., 2019; Saridakis et al., 2019), we use exporting as a proxy for outward internationalisation, defined as 'outward international trade in goods and/or services, conducted either directly or through a third party' (Love & Roper, 2015, p. 29). Previous studies generally support the argument that exporting firms are more productive than non-

exporting firms. This assumption may be explained by either the 'self-selection hypothesis' or the 'learning-by-exporting hypothesis' (Fassio, 2018). The 'self-selection hypothesis' (Bernard & Jensen, 1999; Melitz, 2003) argues that, since the competition level in the export market is higher than that in the domestic market, only more productive firms will have the ability to export. Hence, a higher level of innovation enables firms to gain more access to export markets by increasing their productivity (Monreal-Pérez et al., 2012). Meanwhile, the 'learning-byexporting hypothesis' suggests that exporting firms can become more productive because they are exposed to international markets. Although a great number of studies conclude that firms that introduce innovation are more likely to export (e.g., Roper & Love, 2002; Cassiman et al., 2010; Saridakis et al., 2019), the empirical evidence for learning-by-exporting is relatively inconclusive (Salomon & Shaver, 2005; Bratti & Felice, 2009, 2012; Kiriyama, 2012) and limited, especially in the context of SMEs. Nevertheless, this effect is found to hold (e.g., Salomon & Shaver, 2005; Wagner, 2007; Chang et al., 2013; Bratti & Felice, 2012); hence, we state that firms will be encouraged to introduce innovation due to their participation in international markets and due to the knowledge gained from foreign markets (e.g., Liu & Buck, 2007; Abubakar et al., 2019).

However, although some studies (e.g., Lileeva & Trefler, 2010; Bustos, 2011) found evidence that exporting increases innovation, there is no agreement about which specific type of innovation is mostly affected (Fassio, 2018). It has been implied that innovation activities by exporting firms will differ according to their innovation strategy (Golovko & Valentini, 2014). The review of the previous literature reveals inconsistent results regarding the effect of exporting on different types of innovation, with most studies focusing exclusively on one type. For instance, some studies find a positive association between exporting and product innovation (e.g., Salomon & Shaver, 2005; Iacovone & Javorcik, 2012; Olabisi, 2017), but fail to account for process innovation. Some other studies, in contrast, suggest a positive effect of exporting on process innovation (cf. Damijan et al., 2010) without considering the potential effect on product innovation. These findings, although contributing significantly to our knowledge about the internationalisation-innovation nexus, provide an incomplete picture; therefore, we can argue, for example, that firms may undertake these two types of innovation simultaneously or that some firms may be more prone to engage in either product or process innovation based on the industry in which they operate. Pursuing both innovations simultaneously can be a costly process, especially for SMEs, which tend to lack resources and key know-how such as finance, management and marketing, skilled labour and information (for a review, see Freel, 2000).

The study by Lee et al. (2014) considers both types of innovation and finds that internationalisation in Korean service firms is positively and significantly related to innovation but that the effect is greater on product innovation than on process innovation. A more recent study by Abubakar et al. (2019), using a sample of SMEs in developing countries, finds that there is no association between exporting and product innovation but that there is a negative association between exporting and process innovation. Monreal-Pérez et al. (2012) show that exporting firms are more likely to develop more product innovation than non-exporting firms, while Cooper and Kleinschmidt (1985) show that product innovation enables younger and smaller firms to adapt to foreign market demands. In other words, internationalisation forces firms to update their products to adapt to different market demands and requirements (Silva et al., 2010; Monreal-Pérez et al., 2012). Following this line of argumentation, firms that are engaged in international activities have to adjust their products and improve their quality after entering a foreign market (Alvarez et al., 2013), and the need for introducing product innovation in the international market is stronger for smaller firms than for larger firms (Golovko & Valentini, 2014), given that small firms face significant challenges and liabilities (e.g., smallness and newness) while expanding into foreign markets. It can also be argued, for example, that SMEs are more narrowly focused than larger firms and thus there is a need for them to invest in innovation to adapt their existing products or create new products so that they can compete successfully in the foreign market (Calantone et al., 2004). In addition, product innovation is more important for small firms that are engaged in exporting activities, since it can assist them in mitigating the prices in the export market and overcome the liability of smallness. Bratti and Felice (2012) show that the export status of a firm can positively affect its likelihood of introducing product innovation. They imply that the interaction and communication with foreign buyers can provide the firm with information regarding customers' needs and market demands, which may be translated into product innovation.

The results between process innovation and internationalisation, however, are also revealing. Damijan et al. (2010), for example, find a positive relationship between firm exporting and process innovation. Some researchers, though, argue that SMEs tend to focus their efforts on product innovation rather than process innovation to increase their productivity (e.g., Golovko & Valentini, 2014). Meanwhile, other scholars claim that process innovation, which is based on technological improvements and enhancement in developing production processes, can allow firms to introduce product innovation (Martínez-Ros & Labeaga, 2009). However, according to Bratti and Felice (2012), the probability of introducing process innovation increases for exporting firms only if they first introduce product innovation.

Baldwin and Gu (2004) find that firms that start to export increase their product specialisation and hence their production run, compared with firms that do not export. Importantly, Love and Ganotakis (2013) argue that the stronger competition in the international market puts pressure on exporting firms to improve both their product and their process innovation to operate internationally.

Previous research suggests that product and process development are interlinked. For instance, Hullova et al. (2016) argued that when firms introduce a new process innovation for reducing costs, firms have to adjust their product designs. Previous literature that examine the complementarity between product and process innovation focused on either investigating complementarities-in-use (e.g., Martínez-Ros & Labeaga, 2009) or complementarities-in-performance (e.g., Kotabe & Murray, 1990). The advantages of combining product and process innovation have been highlighted in the complementarities-in-performance stream of literature as it has been argued that combining these two types of innovation will lead to the introduction of new products, cost efficiency and higher returns (Hullova et al., 2016). The complementarity between product and process innovation can be demonstrated by the Industry Life Cycle Theory (Abernathy & Utterback, 1978) in its third phase where the motivation for innovation arise from reducing costs and improving products' quality. For instance, Martínez-Ros (2000) found that firms that introduced process innovation are more likely to introduce product innovation.

Therefore, we argue that complementarity between product and process innovation potentially exists (e.g., Martínez-Ros, 2000; Reichstein & Salter, 2006; Roper et al., 2008; Martínez-Ros & Labeaga, 2009; Van Beers & Zand, 2014; Hullova et al., 2016; Lewandowska et al., 2016), and we propose that outward internationalisation can enable SMEs to gain valuable knowledge and skills related to production, marketing and R&D (Zahra et al., 2009) and to capitalise on opportunities to adapt their products or create new products for foreign markets, as well as adopting new and more efficient methods of production. The preceding discussion leads to the following hypothesis:

H1: Outward internationalisation increases the likelihood of SMEs introducing a single type of innovation or a combination of product and process innovation.

2.2 Inward internationalisation and innovation

Firms can internationalise via two types of operations: outward and inward (Fletcher, 2001; Welch et al., 2007; Hernández & Nieto, 2016). Most of the previous studies, however, tend to

pay more attention to outward international operations, which allow firms to exploit opportunities and obtain knowledge that can be used for their expansion and growth (Pangakar, 2008). However, firms may also internationalise through inward operations, such as importing or contractual collaborations (Welch et al., 2007). Hence, recent studies identify strategic reasons for inward operations, such as their ultimate role in boosting innovation (Nieto & Rodríguez, 2011). Such activities can also aid outward internationalisation as firms could develop important capabilities by learning from foreign sources of knowledge. It is argued that, although imports are rarely viewed as an essential part of firms' internationalisation process, increasing competition forces firms to find ways to lower their costs and gain access to products and knowledge that are not available in their domestic market (Grosse & Fonseca, 2012). Therefore, previous empirical studies show that importing is positively associated with innovation (e.g., Bertschek, 1995; Liu & Buck, 2007; Anderson & Lööf, 2009; Damijan & Kostevc, 2010; Chen et al., 2017).

According to Damijan and Kostevc (2015), firms that have a large number of importing links are more likely to introduce new product or process innovation, which subsequently enhances their productivity and growth. A study by Paunov (2011) shows that firms in Ecuador that are engaged in importing activities are able to influence their product innovation. Meanwhile, Narayanan and Bhat (2009) suggest that a relationship exists between importing technology and firms' R&D. Additionally, Goldberg et al. (2010) find that firms that focus on their domestic markets have the ability to increase their product scope due to the fact that these firms gain access to valuable inputs, which result in the introduction of new products.

Reviewing the existing literature, we argue that SMEs that are engaged in importing are likely to introduce both product and process innovation. To put it differently, importing exposes firms to new processes, since new knowledge tends to be embedded in new machinery (Filippetti et al., 2012; Abubakar et al., 2019), and international suppliers and networks can provide valuable learning opportunities to SMEs (cf. Puthusserry et al., 2020). In addition, firms that are engaged in importing intermediate goods and inputs may have to adjust and advance their production processes. As with the literature on the complementarity between product and process innovation (e.g., Martínez-Ros, 2000; Reichstein & Salter, 2006; Roper et al., 2008; Hullova et al., 2016; Lewandowska et al., 2016; Saridakis et al., 2019), we suggest that inward internationalisation may enable firms to obtain more knowledge from their suppliers, for instance, and therefore have the ability to introduce a combination of product and process innovation. Hence, we hypothesise that:

H2: Inward internationalisation increases the likelihood of SMEs introducing a single type of innovation or a combination of product and process innovation.

2.3 Outward-inward internationalisation and innovation

Although SMEs face significant challenges to internationalise, such as limited experience in internationalisation operations (Cadogan et al., 2012), it can be argued that exposure to foreign markets enables firms to develop capabilities that can enhance their future growth (Lu & Beamish, 2006; Hernández & Nieto, 2016; Puthusserry et al., 2020). Internationalisation via outward operations or inward operations exposes firms to new and diverse ideas, learning opportunities and specialised knowledge that enable them to enhance their ability to develop new skills and introduce innovation (Cheng & Bolon, 1993; Hitt et al., 1997; Chiva et al., 2014). Although the existing work finds that outward and inward internationalisation allow firms to exploit knowledge, gain more advantages and introduce innovation, a gap still exists in the literature regarding the potential impact that can be generated on innovation by implementing both forms of internationalisation operation (i.e., outward and inward) simultaneously (Hernández & Nieto, 2016).

The existing literature suggests that it is critical for firms to gain access to knowledge (Levitt & March, 1998) due to globalisation demands, especially for those firms that are involved in international activities. Outward and inward international operations allow firms to gain access to different and varied types of knowledge and information from a variety of sources. Firms can then develop competitive advantage by leveraging and exploiting diverse sources of knowledge (e.g., March, 1991; Argote, 2011). Studies that examine inward-outward internationalisation focus on investigating how knowledge obtained from inward internationalisation can be used to perform outward internationalisation or vice versa (Hernández & Nieto, 2016). When firms engage in both internationalisation operations at the same time, they may have the ability to develop connections that can create various advantages (Andersson & Lööf, 2012). For example, it is argued that firms can improve their absorptive capacity due to the greater exposure, and to the diversity and complementarity of the accumulated knowledge, hence allowing them to exploit opportunities and discover solutions to their problems and ultimately to achieve better organisational outputs (Zahra & George, 2002; Kostopoulos et al., 2011; Yao et al., 2013). Also, by engaging in both activities, firms can develop experiential learning, which can be vital for enhancing product and process innovation.

According to Hernández and Nieto (2016, p. 297), undertaking both internationalisation activities simultaneously allows firms to increase their 'diversity, relatedness and complementarity of their experiential knowledge', and thereby to increase their absorptive capacity (Yao et al., 2013). The OL literature is generally linked to innovation (Dodgson, 1993). Gomes and Wojahn (2017) argue that firms are able to obtain better competitive advantages and a larger market share when they direct their efforts to innovation. Especially for SMEs, innovation is the key factor that enables them to increase their market share and power (Gunday et al., 2011). Therefore, it can be argued that SMEs that have the ability to learn through internationalisation can be in a better position to detect trends and events in the marketplace and utilise the market opportunities that may emerge. Internationalisation therefore exposes SMEs to various types of knowledge, ideas and learning methods, and to a richer and wider flow of knowledge and information, thus enabling them to innovate. A study by Andersson and Lööf (2012, p. 749) found that Swedish SMEs that are engaged in trade (both exporting and importing) are more likely to 'apply for patents' than firms that are engaged in a single international trade activity (i.e., either exporting or importing). Therefore, we propose that undertaking both internationalisation operations will have a stronger effect on SMEs' innovation. Hence, we hypothesise that the probability of introducing innovation will be stronger when SMEs undertake both internationalisation operations simultaneously (i.e., outward-inward internationalisation).

Given the lack of previous studies that examine the effect of outward and inward internationalisation on different types of innovation, we argue, like the previous literature on the complementarity between product and process innovation, that undertaking both internationalisation operations may allow SMEs to receive a diverse set of knowledge and information, permitting them to introduce a combination of innovation forms. In addition, by applying the OL theory, we also argue that firms can acquire richer knowledge by undertaking both operations, and hence introduce both types of innovation. Based on the above literature, we hypothesise the following:

H3: Outward-inward internationalisation will have a stronger effect on the likelihood of SMEs innovating than a single internationalisation operation.

H4: Outward-inward internationalisation will have a stronger effect on introducing a single type of innovation or a combination of product and process innovation than a single internationalisation operation.

2.4 Theoretical framework

Based on the above literature, Fig (1) represents our conceptual model. The figure shows on the left the internationalisation operation of the firm; whether the firm is an exporter, importer or both an exporter and importer. We expect each type of the internationalisation operation (or a combination of them) to have a positive effect on innovation (H1-H2). Our model allows a disaggregation of the innovation variable, as it is presented on the right side of the figure, and thus allows us to examine whether a combination of outward and inward internationalisation would have a greater estimated effect on overall innovation (H3), and on each type of innovation separately (H4). The data used to measure our variables, the estimated approach adopted, and the empirical findings are presented in the following sections subsequently.

Outward internationalisation Product only innovation Process Inward Innovation innovation internationalisation only Combination of product and process +/-Outward & Inward innovation internationalisation **Controls:** Size Age Legal status Sites Turnover Financial obstacle Intense competition External advice/information R&D tax credit Region Sector

Figure 1: Theorised model.

3. Data and measurements

3.1 Data

This paper uses data from the UK Small Business Survey (BEIS, 2018a) conducted between August 2016 and January 2017 by BMG Research Ltd. The survey is a large-scale telephone survey of 9,248 small business owner-managers (those with up to 249 employees) in the UK, covering firms operating in England, Wales, Scotland and Northern Ireland. Therefore, it can be argued that the survey provides rich information from a large representative sample of UK SMEs (BEIS, 2018a). The Small Business Survey consists of two waves. Information about SMEs' importing activities are not provided in the first wave; therefore, in this paper we use data from the second wave only. Detailed information about the survey method, sampling and instruments can be found in the survey technical report (BEIS, 2018b). Regarding the key variables in this study - exporting, importing and innovation - the survey provides data on (i) whether a firm exports goods and/or services outside the UK, (ii) whether a firm imports goods or services from outside the UK and (iii) whether a firm has introduced a significantly new or improved product or process innovation. These measures are discussed analytically below. The survey also provides information related to firms' characteristics, such as the region, sector, age of the firm, number of employees and turnover. The latter variables are used as controls in our modelling approach.

3.2 Measurements

3.1.1 Dependent variables

We follow previous and most recent empirical studies and measure innovation as the introduction of new products (goods or services) and processes as a proxy for firms' innovative activities (Nguyen et al., 2008; Higón & Driffield, 2010; Higón, 2011; Golovko & Valentini, 2014; Van Beers & Zand, 2014; Fassio, 2018; Abubakar et al., 2019; Saridakis et al., 2019). The survey asks owner-managers the following two questions regarding product and process innovation (BEIS, 2018b, p. 64), respectively:

- 'Has your business introduced any new or significantly improved goods or services in the last three years?'
- 'Has your business introduced any new or significantly improved processes in the last three years?'

The survey does not provide information about the exact year and duration of innovation or whether or not the firm is a serial innovator. Here, we simply create the following two dependent variables to capture the firm's innovative strategic activity:

Overall innovation

Overall innovation is measured through the above two dichotomous scale questions, taking the value of one if the firm introduced product or process innovation individually or jointly and zero otherwise. The survey shows that 41.52 per cent of SMEs introduced innovation, mostly product innovation followed by product and process innovation together (Table 1), while 58.4 per cent did not introduce innovation at all in the last three years.

Different types of innovation

An index is used to capture which types of innovation the firm has introduced. The variable takes the value of one if the firm introduced product innovation, two if the firm introduced process innovation, three if the firm introduced a combination of product and process innovation and zero if the firm did not introduce any innovation. Table 1 shows the distribution of innovation by firm size, suggesting that as the firm size increases, the scale of innovation increases, along with the tendency to carry out process innovation individually or in combination with product innovation. We test whether there are statistically significant differences in the proportions between non-innovative micro, small and medium-sized firms, and find that there are statistically significant differences between micro and small, between micro and medium-sized and between small and medium-sized firms at the 1 per cent level. For the product innovation, process innovation and product and process innovation categories, we also find statistically significant differences, with the sole exception being between small and medium-sized firms (for product innovation prob. = 0.397 and for the combination of product and process innovation prob. = 0.133).

_

¹ We use information from the panel and find that there is a great deal of persistence in non-innovation activity. Nearly 83% and 87% of the firms reported no good/service innovation or process innovation activity, respectively, in either wave. There is less persistence, however, among innovative firms, with nearly half of the firms that reported innovation activity in wave 1 reporting innovation in wave 2. This is because innovation is costly, and SMEs especially have restricted access to financial resources to allocate for innovation. Hence, if a firm admitted innovation activity in wave 1 but not in wave 2, it suggests that those SMEs have been free of innovation for three years or more since the last innovation.

Table 1: Innovation types by firm size, %.

	All SMEs	Micro firms (<10 employees)	Small firms (10-49 employees)	Medium firms (50-249 employees)		
No innovation	58.48	64.17	52.21	46.89		
Product innovation only	19.91	18.54	21.43	22.69		
Process innovation only	6.40	5.45	7.10	8.99		
Product and process innovation	15.21	11.85	19.26	21.43		
Observations	8,171	4,811	2,170	1,190		

Note: Percentages may not add up to 100 due to rounding.

3.1.2 Independent variables

Outward internationalisation (i.e., exports)

Similarly, following the previous literature, we use exports as a proxy for internationalisation (e.g., Salomon & Shaver, 2005; Higón & Driffield, 2010; Ganotakis & Love 2011; Liang et al., 2012; Serra et al., 2012; Boehe, 2013; Golovko & Valentini, 2014; Idris & Saridakis, 2018; Abubakar et al., 2019; Saridakis et al., 2019). The survey asks owner-managers the following question:

• 'In the past 12 months did your business export any goods and/or services outside the UK?' (BEIS, 2018b, p. 36).

This construct takes the value of one if the firm exports goods and/or services outside the UK and zero otherwise.² The survey shows that 21.67 per cent of SMEs in the UK exported goods and/or services outside the UK. Also, 59.29 per cent of the exporting SMEs had introduced innovation in the last three years. In addition, 20.76 per cent of micro firms that export introduced a combination of product and process innovation, whereas 29.70 per cent of small firms and 28.77 per cent of medium-sized firms did so.

Inward internationalisation (i.e., imports)

Based on the previous literature (e.g., Hernández & Nieto, 2016; Abubakar et al., 2019), we use imports as a proxy for inward internationalisation. The survey asks the participants to respond to the following question:

• 'In the past 12 months, have you directly imported any goods or services from outside the UK?' (BEIS, 2018b, p. 38).

² Again, we use information from the panel and find that there is much persistence in both exporting firms (79%) and non-exporting firms (96%) between waves 1 and 2.

³ The survey asks owner-managers whether the imports were from the EU or outside the EU. However, this differentiation is beyond the scope of this paper, so we group them into one category.

We constructed a variable taking the value of one if the firm imported goods or services from outside the UK and zero otherwise. The survey shows that 22.82 per cent of SMEs imported goods or services from outside the UK. Moreover, 58.97 per cent of the importing firms had introduced innovation in the last three years and 22.63 per cent of the importing micro firms had introduced a combination of product and process innovation (followed by 28.11 per cent of the importing medium-sized firms and 25.81 per cent of the importing small firms).

Outward and inward internationalisation (i.e., exports and imports)

To capture the different internationalisation operations of firms, we follow recent work (e.g., Seker, 2009; Hernández & Nieto, 2016) and distinguish between firms according to their internationalisation operations. In particular, we create an index to distinguish between firms that are (i) only involved in outward internationalisation (i.e., exporting), (ii) only involved in inward internationalisation (i.e., importing), (iii) involved in both international operations (i.e., exporting and importing) simultaneously and (iv) not involved in any international operations (reference category). Table 2 shows that about 20 per cent of medium-sized firms are engaged in exporting and importing activities, followed by 18.32 per cent of small firms and 7.85 per cent of micro firms. We also test whether there are statistically significant differences in the proportions between different-sized bands with different internationalisation operations. The results show that, for exporting firms, there is no statistically significant difference between micro and small firms that export only (prob. = 0.420), between micro and medium-sized firms that export only (prob. = 0.565) or between small and medium-sized firms that export only (prob. = 0.949). Moreover, we find that the difference between micro and small firms that import only is statistically insignificant (prob. = 0.310). As for importing and exporting firms, the results show that all coefficients are statistically different, with the sole exception of between small and medium-sized firms (prob. = 0.227).

Table 2: Firms' internationalisation operations by firm size, %.

	All SMEs Micro firms (employees)		Small firms (10-49 employees)	Medium firms (50-249 employees)		
No international operations				-		
(i.e., no trade)	67.97	73.59	61.73	56.65		
Outward operations (i.e.,						
exports) only	9.21	8.98	9.58	9.51		
Inward operations (i.e.,						
imports) only	10.41	9.58	10.37	13.80		
Outward-inward operations						
(i.e., both exports and imports)	12.41	7.85	18.32	20.03		
Observations	8139	4790	2161	1188		

Note: Percentages may not add up to 100 due to rounding.

3.1.3 Control variables

In our specification, we control for several variables, such as the firm's age (e.g., Hansen, 1992; Abubakar et al., 2019), measured as the number of years since the firm started operating, the firm's turnover, its size (Higón, 2011), the number of sites it has (e.g., Roper & Love, 2002) and its legal status (Higón & Driffield, 2010). Following previous work (e.g., Andersson & Lööf, 2009; Abubakar et al., 2019), we also control for the firm's external environmental factors, specifically its ability to obtain financial resources, the market competition and the networking linkages (for the latter, see Kingsley & Malecki, 2004; Rogers, 2004; Saridakis et al., 2019). It is argued, for example, that access to finance is often a major obstacle for small firms to carry out innovative activities. Moreover, we control if the firm has received R&D tax credits in the past three years. Finally, a general argument in the neo-Schumpeterian literature is that the characteristics of a particular sector or industry may influence its innovative activities (Andersson & Lööf, 2012); hence, we control for the sector effects along with the regions. Table A1 in the Appendix presents the definitions of the variables used in this paper. In addition, also in the Appendix, we present the summary statistics of the key variables (Table A2) and the correlation matrices (Tables A3-A4).

4. Empirical model and results

4.1 Empirical model

To test the association between internationalisation and innovation, we use two statistical approaches. First, since overall innovation takes only two possible values (1 if the firm introduced innovation and 0 otherwise), we use a probit regression (for a discussion, see Gujarati 1995: 552-570) to examine the association between internationalisation operations

(i.e., exporting, importing and both exporting and importing) and innovation.⁴ Our model can be written as follows:

$$I_{j} = \begin{cases} 0 & \text{if } I_{j}^{*} \leq 0 \\ 1 & \text{if } I_{j}^{*} > 0 \end{cases}$$
 (1)

$$I_i^* = X_i b_i + E X_i \vartheta + I M_i \vartheta + E X I M_i \gamma + e_i$$
 (2)

where I_j^* denotes the latent variable and EX, IM and EXIM are the indicator variables for whether the firm has exported, imported or both (i.e. exported and imported), respectively. X is the vector of firm characteristics for firm j. b, ϑ , δ and γ are the parameters to be estimated. In addition, following the previous literature (e.g., Wagner, 2002; Yasar & Rejesus, 2005; Saridakis et al., 2019), we apply propensity score matching techniques (Rosenbaum & Rubin, 1983) to reduce the potential biases and allow for stronger causal inferences between the internationalisation operations and innovation.

Second, when the innovation variable is disaggregated to capture a single (i.e., either product innovation or process innovation) or combined type of innovation (i.e., both product and process innovation), the above model is re-estimated using a multinomial logit model (the base category is no innovation), which is a generalisation of the binary logit model (Brooks, 2008). We test for the independence of irrelevant alternatives (IIA) using both the Hausman and the Small–Hsiao test. Both tests suggest that the IIA has not been violated. Our dependent variable in this case is a categorical and unordered variable, j = 4, and can be written as follows:

$$I_{j} = \begin{cases} 0 & \text{if} \quad I \neq j \\ 1 & \text{if} \quad I = j \end{cases} \tag{3}$$

For both models, we present the marginal effects to assist with the interpretation of the results and estimate the particular effects for each category (in the case of the multinomial logit model).

20

⁴ We also use a logit model; however, the results are similar and thus are not reported here.

4.2 Empirical results

4.2.1 The relationship between internationalisation operations and innovation

In Table 3, we present the marginal effect of the probit analysis of the relationship between internationalisation operations and innovation for all SMEs.⁵ The results presented in Model 1 of Table 3 show that compared with no trading, exporting only increases the probability of introducing innovation by 12.3 percentage points, while importing only increases the probability of introducing innovation by 17.3 percentage points. In addition, the results show that carrying out both exporting and importing activities increases the likelihood of introducing innovation by 23.2 percentage points. To examine the robustness of this finding, we also use propensity score matching techniques (see Cameron & Trivedi, 2005, for a technical discussion) that account for potential endogeneity. Using a model that allows multiple nominallevel treatments, the results are found to be consistent with those reported in Table 3 and thus are not discussed here (see Model 3, Table 5A in the Appendix). Using the Wald test (see Judge, Griffiths, Hill, Lutkepohl & Lee, 1985), we determine whether these coefficients are statistically different from each other. The results show that the coefficient of exporting is statistically significantly different from the coefficient of importing ($x^2(1) = 4.49$ and prob. = 0.034) and the one reported for the combination of exporting and importing ($x^2(1) = 17.72$ and prob. = 0.001). Moreover, the results show that the coefficient of importing is statistically significantly different from the coefficient of the combination of exporting and importing $(x^2(1) = 4.67 \text{ and prob.} = 0.030)$. Hereafter, we restrict the sample to those SMEs that import. The results in Model 2 of Table 3 show that compared with importing, exporting increases the likelihood of innovation by 11.7 percentage points. In addition, we find that the combination of exporting and importing increases SMEs' innovation by 22.2 percentage points compared with importing. Finally, the results show that the coefficients of exporting and exporting and importing together are statistically significantly different from each other ($x^2(1) = 16.37$ and prob. = 0.001). Overall, the results in Table 3 confirm our H1-H3 which state that each internationalisation operation will be positively associated with innovation, and that outward-

-

⁵ We test if exporting is associated with innovation, and the results confirm previous literature and so are not discussed here (exporting increases the likelihood of introducing innovation by 13.9 percentage points compared to non-exporting firms). Similar results are obtained for importing firms compared to non-importing firms (18 percentage points). We also use the propensity score matching techniques that account for potential endogeneity between exporting and innovation and between importing and innovation. We apply the nearest neighbour estimator, and the results show that for firms that are engaged in exporting (importing), exporting (importing) has caused the probability of introducing innovation to be 16 percentage (12.9 percentage) points higher than it would have been otherwise (see Table A5 in the Appendix).

inward internationalisation will have a stronger effect on the likelihood of SMEs' innovation than a single internationalisation operation.

Table 3: The association between internationalisation operations and overall innovation (all SMEs), *probit estimates (ME and standard errors)*.

	Overall innovation				
	Model (1)	Model (2)			
Internationalisation operations (Base category: No trade)		. ,			
Exports Only	0.123***				
	0.021				
Imports Only	0.179***				
	0.019				
Exports & imports	0.232***				
•	0.020				
Internationalisation operations (Base category: Imports)					
Exports only		0.117***			
		0.021			
Exports & imports		0.222***			
		0.020			
Key independent variables					
Firm size	0.036***	0.036***			
	0.007	0.007			
Financial obstacle	0.106***	0.109***			
	0.015	0.016			
Intense competition	0.054***	0.051***			
1	0.011	0.012			
External advice/information	0.164***	0.165***			
v	0.013	0.013			
R&D Tax credit	0.209***	0.219***			
	0.027	0.029			
Other controls	Yes	Yes			
log Likelihood	-4937.8824	-4373.7321			
x^2 (degrees of freedom)	1170.45(45)	1074.35(44)			
Observations	8139	7292			
Wald test (x^2)					
Exports only = Imports only = Exports & imports	17.76***				
	0.001				
$Exports \ only = Imports \ only$	4.49**				
	0.034				
$Exports \ only = Exports \ \& \ imports$	17.72***	16.37***			
	0.001	0.001			
Imports only = Exports & imports	4.67**				
1 1	0.030				

Notes: All the models include control variables (full results are available upon request). The standard errors of the coefficients are in *italics*.

As a robustness check, we also estimate a logit model, and the results are found to be similar to those reported here. Values in *italics* are standard errors. *** p < 0.01, ** p < 0.05, * p < 0.1.

4.2.2 The relationship between internationalisation operations and different types of innovation

Table 4 presents the marginal effect of the multinomial logit analysis of the relationship between internationalisation operations and different types of innovation for all SMEs (we exclude the marginal effect of the base category, which is no innovation). The results show that firms that are engaged in exporting increase the probability of introducing product innovation only (Model 1a, Table 4) by 6.6 percentage points and the combination of product and process innovation (Model 1c, Table 4) by 5.3 percentage points. However, the results show that exporting is not associated with process innovation. Therefore, although the results continue to support our *H1*, they indicate that outward internationalisation increases the likelihood of SMEs introducing a product innovation only, and a combination of product and process innovation. Moreover, we find that importing increases the probability of introducing product innovation and a combination of product and process innovation by 7.5 percentage points and 7.3 percentage points, respectively (Model 2a, 2c, Table 4). Again, we find that importing is not associated with process innovation. Hence, the results support our *H2* indicating that inward internationalisation increases the likelihood of SMEs introducing a product innovation and a combination of product and process innovation, but not a process innovation.

When differentiating between firms according to their internationalisation operations, the results suggest that the probabilities of introducing product innovation only and a combination of product and process innovation are both associated with engaging in different international operations. More specifically, the results (Models 3a, 3c, Table 4) show that, compared with no trading at all, the combination of exporting and importing increases the probability of introducing product innovation only and a combination of product and process innovation by 10.4 percentage points and 9.2 percentage points, respectively. Importing only, on the other hand, only increases the probability by 6.2 percentage points for product innovation and 8.1 percentage points for a combination of product and process innovation. Meanwhile, exporting only is found to increase the probability of introducing product innovation by 4.8 percentage points and the probability of a combination of product and process innovation by 5.4 percentage points. Overall, the results show that outward-inward internationalisation has a stronger effect on introducing a combination of product and process innovation than a single internationalisation operation.

We test whether the above coefficients are statistically significantly different from each other. The results show that, for the coefficients associated with product innovation (Model 3a, Table 4), both exporting only and importing only are statistically significantly different from

the combination of exporting and importing ($x^2(1) = 14.65$ and prob. = 0.001; and $x^2(1) = 6.26$ and prob. = 0.012, respectively); while, for the coefficients associated with product and process innovation (Model 3c, Table 4), the results show that importing only is not statistically significantly different from the combination of exporting and importing ($x^2(1) = 1.94$ and prob. = 0.164). Hence, our *H4* seems to be partially supported.

When restricting the sample to those SMEs that import only, the results support the previous findings. The results (Model 4a, 4c, Table 4) show that exporting only increases the probability of introducing product innovation by 4.5 percentage points, and by 5 percentage points for introducing a combination of product and process innovation. In addition, the results show that the combination of exporting and importing increases the probability of introducing product innovation by 9.8 percentage points and the probability of introducing product and process innovation by 8.4 percentage points. We test whether these coefficients are statistically significantly different from each other, and the results suggest that they are indeed ($x^2(1) = 13.39$ and prob. = 0.001 for Model 4a, $x^2(1) = 11.53$ and prob. = 0.001 for Model 4c).

Table 4: The association between internationalisation operations and different types of innovation (all SMEs), *mlogit estimates (ME and standard errors)*.

Type of innovation:	Product innovation			Process innovation				Product and process innovation				
	Model (1a)	Model (2a)	Model (3a)	Model (4a)	Model (1b)	Model (2b)	Model (3b)	Model (4b)	Model (1c)	Model (2c)	Model (3c)	Model (4c)
Exports	0.066***				0.001				0.053***			
_	0.011				0.007				0.009			
Imports		0.075***				0.008				0.073***		
		0.010				0.006				0.009		
Internationalisation operations (base category: no trade)												
Exports only			0.048*** 0.015				0.002 0.009				0.054*** 0.012	
Imports only			0.062*** 0.014				0.012 0.008				0.081*** 0.011	
Exports and imports			0.104*** 0.014				0.004 0.009				0.092*** 0.011	
Internationalisation opera category: imports)	tions (base											
Exports only				0.045*** 0.015				0.003 0.009				0.050*** 0.012
Exports and imports				0.098*** 0.014				0.006 0.009				0.084*** 0.011
Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
log Likelihood	-8288.429	-8219.1534	-819.8806	-7200.619	-8288.429	-8219.1534	-819.8806	-7200.619	-8288.429	-8219.1534	-819.8806	-7200.619
x²(degrees of freedom)	1358.58(129)	1420.54(129)	1465.09(135)	1320.48(132)	1358.58(129)	1420.54(129)	1465.09(135)	1320.48(132)	1358.58(129)	1420.54(129)	1465.09(135)	1320.48(132)
Observations	8171	8139	8139	7292	8171	8139	8139	7292	8171	8139	8139	7292
Wald test (x²) Exports only = Imports only = Exports & Imports Exports only = Imports only			15.40*** 0.001 1.94 0.163								6.57** 0.037 4.83** 0.027	
Exports only = Exports & Imports Imports only = Exports & Imports			0.763 14.65*** 0.001 6.26** 0.012	13.39*** 0.001							0.027 13.07*** 0.001 1.94 0.164	11.53*** 0.001

Notes: All the models include control variables (full results are available on request). The standard errors of the coefficients are in *italics*.

Values in *italics* are standard errors. *** p < 0.01, ** p < 0.05, * p < 0.1.

4.2.3 Robustness check

4.2.3.1 Results by firm size

Although the results are not presented here due to brevity, we perform a robustness check and differentiate between firms according to their size-bands (i.e., micro, small and medium). We first test the association between different internationalisation operations and innovation. Some of these findings are interesting and thus discussed here. We find that all internationalisation operations increase the probability of introducing innovation in micro and small firms; however, for medium-sized firms, exporting only is not associated with innovation. For example, we find that, compared with no trading, exporting only in micro firms increases the probability of introducing innovation by 10.5 percentage points, while importing only increases the probability of innovation by 19.6 percentage points. However, the results show that a combination of exporting and importing has a stronger effect on the probability of introducing innovation in micro firms (increasing the probability by 26.8 percentage points). We find that the estimated coefficients are statistically significantly different from each other.⁶ Similar results are obtained for small firms, that is, 18.8 percentage points for exporting only, 18 percentage points for importing only and 23.1 percentage points for exporting and importing.⁷ However, for medium-sized firms, the results show that, isolating the exporting activity, exporting only is not associated with innovation. In contrast, operating an importing activity only is shown to increase the probability of introducing innovation by 8.8 percentage points. However, when both exporting and importing activities are performed together, they are found to increase the probability of innovation by 9.5 percentage points.⁸

Also, we perform similar tests to examine the association between different internationalisation operations and different types of innovation. The results show that, in micro firms for example, different types of internationalisation operations increase the likelihood of introducing product innovation and a combination of product and process innovation. For instance, the results suggest that exporting only in micro firms increases the probability of introducing product innovation by 4.1 percentage points and product and process innovation by 4.2 percentage points. Importing only, however, increases the probability of

⁶ The results show that exporting only is statistically significantly different from importing ($x^2(1) = 6.85$ and prob. = 0.008) and from exporting and importing ($x^2(1) = 19.19$ and prob. = 0.001). In addition, importing is statistically significantly different from exporting and importing ($x^2(1) = 3.95$ and prob. = 0.046).

⁷ We also test whether these coefficients are statistically significantly different from each other, and the results suggest that they are not.

⁸ We test whether these two coefficients are statistically significantly different from each other, and the results show that they are not ($\chi^2(1) = 0.01$ and prob. = 0.903).

introducing product innovation by 7 percentage points and product and process innovation by 8.8 percentage points.

In addition, the results show that the combined internationalisation operation (exporting and importing) has a stronger effect on introducing product innovation (increasing the probability by 11.5 percentage points) and a combination of product and process innovation (increasing the probability by 9 percentage points). Using the Wald test, we determine whether these coefficients are statistically significantly different from each other. We found that, in the product innovation equation, exporting is statistically significantly different from both exporting and importing activities ($x^2(1) = 13.66$ and prob. = 0.001) and from importing ($x^2(1) = 3.25$ and prob. = 0.071), as well as that importing is also different from both exporting and importing activities ($x^2(1) = 4.11$ and prob. = 0.042). Similar results are obtained for small firms, where a combination of exporting and importing has the strongest effect on introducing product innovation and a combination of product and process innovation (i.e., 11.3 percentage points and 9.8 percentage points, respectively). On the other hand, the results for medium-sized firms show that only a combination of exporting and importing increases the likelihood of introducing a combination of product and process innovation (by 7 percentage points), compared to no trading at all.

4.2.3.2 Results for restricted sample

As mentioned earlier, our data provides no information about the year of the firm's introduced innovation and how often the firm innovates. However, we combine information from the survey on when the firm started exporting for the first time. We find that less than 10 per cent of firms reported exporting activity in the last year have up to 3 years exporting experience. When we run the innovation model excluding the firms with 3 years or less experience in exporting, we still find that exporting activity increases the probability of innovation. There is no information about when the firms started importing goods to repeat this robustness check for the relationship between importing and innovation. To examine these relationships further, we use information from the earlier wave, and generate a sample of 2,967 firms that a) have reported innovation activity at the time of the second wave only, (i.e., responded positively to the innovation question in wave 2 but not in wave 1) as well as b) firms that have reported no innovation activity in both waves. The former group (a) consists of innovators firms and the second group (b) consists of non-innovators firms. The results from this model suggest that firms with importing activity and a combination of exporting and importing activities are more

likely to innovate than firms with no trade at all, but this was not the case for exporting that is found not to be statistically different for no trade (see Figure B1 in Appendix; full results available upon request). In particular, we find that these variables are strongly related to product innovation.⁹

5. Discussion

This paper examines the relationship between different internationalisation operations (i.e., outward internationalisation, inward internationalisation and outward-inward internationalisation) and different types of innovation in SMEs. Such studies are rare in the IB field that have examined both activities and their impact on innovation (cf. Bagheri et al., 2019; Puthusserry et al., 2020). Our results show that there is a direct and positive relationship between outward internationalisation (i.e., exporting) and innovation, and between inward internationalisation (i.e., importing) and innovation in all SMEs.

Our results therefore provide support for the limited but growing number of studies that highlight the importance of learning by exporting (e.g., Liu & Buck, 2007; Lileeva & Trefler, 2010; Filippetti et al., 2012; Amodóvar et al., 2014; Love & Roper, 2015; Hernández & Nieto, 2016; Chen et al. 2017; Damijan et al., 2017). Moreover, our results are consistent with some previous studies that examine the effect of importing on innovation (e.g., Liu & Buck, 2007; Fritsch & Görg, 2015; Hernández & Nieto, 2016; Chen et al., 2017; Abubakar et al., 2019).

Most importantly, when we examine the relationship between outward-inward internationalisation, in the form of exporting and importing, and innovation in all SMEs and in different sizes of SMEs, we find interesting results. Our results suggest that engaging in a single individual international operation (either exporting or importing) or in both international operations simultaneously increases innovation in SMEs - however the effect is found to be larger in the magnitude for firms that perform both operations at the same time. These results are confirmed in the full-sample analysis and in the sub-sample analysis for micro and small firms. As for medium-sized firms, the results show that, compared with not trading at all, exporting only is not associated with innovation. The findings suggest that only engaging in

_

⁹ We also split this sample into young firms (5 years or less) and older ones (6 years and above); for the former group, the firms reporting innovation are likely to be firms with no prior history of innovation (i.e., these firms have reported no innovation for the last three years in the previous survey). For young firms, importing activity is found to be associated with innovation, whereas for older ones both importing and a combination of exporting and importing are found to increase the probability of innovation compared to firms with no trade activity.

importing activities, or engaging in both internationalisation operations simultaneously, have a significant association with innovation in medium-sized firms.

Moreover, when we differentiate between different types of innovation (i.e., product innovation, process innovation, product and process innovation), the results provide evidence on the association between exporting and the introduction of product innovation and a combination of product and process innovation. Hence, our results are consistent with previous studies that examine the effect of exporting on product innovation (e.g., Higón & Driffield, 2010; Bratti & Felice, 2012; Iacovone & Javorcik, 2012; Monreal-Pérez et al., 2012; Lee et al., 2014; Olabisi 2017). However, we do not find evidence of the association between exporting and process innovation (e.g., Damijan et al., 2010; Higón & Driffield, 2010; Abubakar et al., 2019); we argue that this may be due to the fact that we introduce a combination of product and process innovation, which isolates the pure individual effects.

The sub-sample analysis reveals some interesting results as well. The results show that micro and small firms engaged in outward internationalisation (i.e., exporting) have a higher probability of introducing product innovation and a combination of product and process innovation. However, we find that, for medium-sized firms, exporting is not associated with any type of innovation. We argue, alongside the previous literature, that the need for product innovation is stronger for smaller firms than for larger firms (Golovko & Valentini, 2014). Since small firms face more resource constraints compared with larger firms, they tend to invest to adapt their existing products or introduce new products to the market so that they can continue operating and competing successfully in the export market. In conclusion, our findings suggest that learning and acquiring knowledge through exporting is potentially important for innovation in SMEs (Kafouros et al., 2008; Higón & Driffield, 2010; Altomonte et al., 2014; Amodóvar et al., 2014).

In addition, we find evidence on the association between importing and product innovation and between importing and product and process innovation in all SMEs, as well as in the split analysis between different-sized firms. Our results are consistent with the previous literature that finds a positive effect between importing and product innovation (e.g., Blind & Jungmittag, 2004; Liu & Buck, 2007; Goldberg et al., 2010; Paunov, 2011; Abubakar et al., 2019). We also find evidence on the association between importing and a combination of product and process innovation. Although previous studies (e.g., Bertschek, 1995; Damijan & Kostevc, 2010) find evidence for their individual effects, we argue that the acquisition of knowledge obtained from importing is an important channel for international knowledge

spillovers (e.g., Coe & Helpman, 1995; Keller, 1998), and therefore it can enable firms to engage in introducing both types of innovation at the same time.

When differentiating between firms according to their size bands, we obtain similar results. The findings show that micro, small and medium-sized firms that are engaged in importing activities have a higher probability of introducing product innovation and a combination of product and process innovation. We argue alongside previous studies (e.g., Paunov, 2011) that the gains from importing intermediate goods or inputs allow firms to obtain new and previously unavailable different inputs that allow for better production methods or different final outputs.

Lastly, we find that outward-inward internationalisation operation has a stronger effect on introducing product innovation and a combination of product and process innovation in the sample of all SMEs and in the different-sized sub-sample estimates. We argue that our results are consistent with the literature on organisational learning in its emphasis on the role of knowledge acquisitions for the development of absorptive capacity (e.g., Cohen & Levinthal, 1990; Zahra & George, 2002; Eriksson & Chetty, 2003). It is suggested that firms that look beyond their national borders have the ability to acquire internationalisation and market knowledge (Eriksson et al., 1997). In this context, our results are consistent with the existing research that considers the effect of both outward and inward international operations simultaneously (e.g., Hernández & Nieto, 2016). Carrying out both internationalisation operations at the same time helps firms to combine the diverse knowledge in a more comprehensive way, therefore generating complementary knowledge that can increase the opportunity for learning and result in better forms of innovation.

6. Conclusion, implications and limitations

Internationalisation studies traditionally focus on analysing outward operations, with most scholars, until recently, ignoring the critical role of inward operations (Quintens et al., 2006; Westney, 2020). Moreover, although research on outward and inward operations have started to receive more attention from scholars, it remains an area of study in which many research and policy questions await answers (Hernández & Nieto, 2016; Westney, 2020). In this paper, we contribute to the previous literature on the association between outward and inward internationalisation operations in firms' innovation in the context of SMEs. We follow previous recent studies (e.g., Hernández & Nieto, 2016; Gerschewski et al., 2018) and apply the organisational learning theory, which stresses the role of knowledge in developing absorptive capacity (Zahra & George, 2002; Eriksson & Chetty, 2003). It is suggested that firms that

operate beyond their national borders gain different types of knowledge, such as internationalisation knowledge, market knowledge and technological knowledge (Eriksson et al., 1997; Fletcher & Harris, 2012), and such knowledge can be vital important for developing innovation. Therefore. this research shows that undertaking outward-inward internationalisation operations simultaneously allows firms to gain diverse knowledge that increases their opportunities for experiential learning and aiding their absorptive capacity, which can ultimately enhance their innovation activities. By gaining diverse knowledge can have an important influence on SMEs' innovation (Hartman et al., 1994). In particular, this paper adds to the development and enhancement of our understanding of the role and the impact of conducting outward and inward internationalisation simultaneously on firms' innovation, which is relatively limited discussed so far in both the IB and the small business literature.

From an academic point of view, this research contributes to the ongoing debate regarding inter-connected internationalisation strategies and their potential effect on firms (Hernández & Nieto, 2016). This paper advances our understanding of the importance of different types of internationalisation operations and their effect on SMEs' innovation. Our argument points towards the idea that sharing related and diverse knowledge and information through different internationalisation operations can potentially increase firms' absorptive capacity in the form of innovation. The results show that each internationalisation operation itself has a different effect on innovation; however, the effect of combined internationalisation operations has a greater effect on innovation in SMEs. These findings have important implications for the wider literature on internationalisation as it provides important insights on the fundamental role of diverse and variety of sources of knowledge for resource constrained SMEs to develop innovation. The extant process view of internationalisation has either focused on the outward internationalisation aspect as a form of gaining market knowledge or only on inward internationalisation (cf. Johanson & Vahlne, 1977, 1990). This study provides a unique and much needed multidimensional - holistic view of both activities and explicate their role in innovation. By engaging in both activities, SMEs can simultaneously explore and exploit external sources of knowledge for developing competitive advantage (cf. March, 1991; Argote, 2011; Puthusserry et al., 2020). The engagement in both activities give access to diverse networks of knowledge and provides double loop learning opportunities to firms, and through such learning SMEs can develop innovative capabilities, which further enables their internationalisation process (e.g., Johanson & Vahlne, 2009; Puthusserry et al., 2020). Overall, we contribute to the extant literature and particularly focus on both outward and inward activities and their effect on SMEs' innovation, which to date has been underexplored in the

extant literature (Welch & Luostarinen, 1993; Hernandez & Nieto, 2016; Li et al., 2017; Bagheri et al., 2019; Westney, 2020).

From a managerial and policy makers' point of view, this research suggests that owner-managers of small firms should not limit their consideration to only the positive effect of undertaking a single type of international operation; rather, firms' innovation is strongly affected by undertaking both outward and inward international operations simultaneously. The specific and diverse knowledge flows from these two operations may lead to a higher probability of introducing innovation, more specifically product innovation and a combination of product and process innovation. Hence, it can be suggested that our research is of importance for owner-managers of SMEs, because, despite their limited resources, these firms can benefit from different international operations, especially where intangible resources, such as knowledge flows, are fundamental to their operations. As for policy makers, this study argues, alongside previous studies (e.g., Korhonen et al., 1996; Hernández & Nieto, 2016), that governments should promote programmes that encourage not only entry into foreign markets but also international sourcing through importing.

Our research has some limitations that may provide interesting lines for future research. First, due to the data limitation and constraints of the measures used, our conclusions should be interpreted with caution; we can only provide theoretical justification for the claim that organisational learning increases firms' absorptive capacity when outward and inward internationalisation operations are performed simultaneously. The association should be better explored in a time-series or a panel data framework that allows for studying dynamics and exploring causality between the variables. Further research may include empirical measures of the organisational learning generated from international operations and investigate whether this generates an effect on the association between outward-inward internationalisation and innovation via moderation and/or mediation. Second, our measure of international operations is limited to the propensity to export to and import from international markets. Further research may consider including other measures of internationalisation, such as the intensity of exporting and importing, to test the association for highly internationalised SMEs. To this end, we call for more theoretical and empirical studies in the small business innovation and international activity and their connections with international performance. Finally, further research should differentiate between novel innovation and incremental innovation to

determine the association between different internationalisation operations and these types of innovation.¹⁰

To conclude, this paper examines the association between different internationalisation operations and innovation in SMEs. Given the gaps in the previous literature, this paper does not limit its analysis to examining a single type of internationalisation operation. Contrary to previous empirical studies, this paper examines the effect of outward, inward and outward-inward internationalisation operations. Moreover, we reach beyond this distinction to examine the above association for different types of innovation introduced by SMEs (i.e., product, process, and product and process innovation). Most importantly, this paper examines this association for the first time in different-sized SMEs. It highlights the role of organisational learning and the knowledge flow generated by engaging in both internationalisation operations for product innovation and a combination of product and process innovation. Accordingly, this paper contributes to the previous literature by providing empirical evidence indicating that SMEs that are engaged in outward and inward internationalisation operations simultaneously are able to take greater advantage of the acquired knowledge flow and information and have a higher probability of introducing innovation.

Funding: This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

_

¹⁰ We would like to thank reviewer 2 for pointing this out.

References

- Abernathy, W.J., Utterback, J.M., 1978. Patterns of industrial innovation. Technology Review 80(7), 41–47.
- Abubakar, Y.A., Hand, C., Smallbone, D., Saridakis, G., 2019. What specific modes of internationalization influence SME innovation in Sub-Saharam least developed countries (LDCs)? Technovation 79, 56–70.
- Abubakar, Y.A., Mitra, J., 2009. Innovation performance in European regions: comparing manufacturing and services ICT subsectors. International Journal of Entrepreneurship and Innovation Management 11(2), 156–177.
- Acs, Z.J., (2002). Innovation and the growth of cities. Cheltenham: E. Elgar.
- Adomako, S., Opoku, R.A., Frimpong, K., 2017. The moderating influence of competitive intensity on the relationship between CEOs' regulatory foci and SME internationalization. Journal of International Management 23(3), 268–278.
- Aldrich, H., Auster, E.R., 1986. Even dwarfs started small: Liabilities of age and size and their strategic implications. Research in Organizational Behavior 8, 165–198.
- Al-Laham, A., Tzabbar, D., Amburgey, T.L., 2011. The dynamics of knowledge stocks and knowledge flows: innovation consequences of recruitment and collaboration in biotech. Industrial and Corporate Change, 20(2), 555–583.
- Alegre, J., Chiva, R., 2008. Assessing the impact of organizational learning capability on product innovation performance: An empirical test. Technovation 28, 315–326.
- Alegre, J., Pla-Barber, J., Chiva, R. Villar, C., 2012. Organisational learning capability, product innovation performance and export intensity. Technology Analysis and Strategic Management 24(5), 511–526.
- Altomonte, C., Aquilante, T., BéKés, G., Ottaviano, G., 2014. Internationalization and innovation of firms: Evidence and policy. Centre for Economic Performance, Special paper No. 32, 1–43.
- Alvarez, R., Faruq, H., López, R., 2013. Is pervious export experience important for new exports? Journal of Development Studies 49(3), 426–441.
- Amodóvar, P., Saiz-Briones, J., Silverman, B.S., 2014. Learning through foreign market participation: The relative benefits of exporting, importing, and foreign direct investment. The Journal of Technology Transfer 39, 915–944.
- Andersson, M., Lööf. H., 2009. Learning-by-exporting revisited: The role of intensity and persistence. Scandinavian Journal of Economics 111(4), 893–916.
- Andersson, M., Lööf. H., 2012. Small business innovation: Firm level evidence from Sweden. Journal of Technology Transfer 37, 732–754.
- Argote, L., 2011. Organizational learning research: Past, present and future. Management Learning 42(4), 439–446.
- Argote, L., Miron-Spektor, E., 2011. Organizational learning: From experience to knowledge. Organization Science 22(5), 1123–1137.
- Argyris, C., Schon, D. 1978. *Organizational learning: A theory of action perspective*. Reading, MA: Addison-Wesley.
- Azari, M.J., Madsen T.K., Moen, O., 2017. Antecedent and outcomes of innovation-based growth strategies for exporting SMEs. Journal of Small Business and Enterprise Development 24(4), 733-752.
- Bagheri, M., Mitchelmore, S., Bamiatzi, V., Nikolopoulos, K., 2019. Internationalization orientation in SMEs: The mediating role of technological innovation. Journal of International Management 25(1), 121–139.
- Baldwin, J.R., Gu, W., 2004. Trade liberalization: Export market participation, productivity growth, and innovation. Oxford Review of Economic Policy 20(3), 372–392.

- Bernard, A.B., Jensen, J.B., 1999. Exceptional exporter performance. Cause, effect or both? Journal of International Economics 47(1), 1–25.
- Bertrand, O., 2011. What goes around, comes around: Effects of offshore outsourcing on the export performance of firms. Journal of International Business Studies 42(2), 334–344.
- Bertschek, I., 1995. Product and process innovation as a response to increasing imports and foreign direct investment. The Journal of Industrial Economics 341–357.
- Blind, K., Jungmittag, A., 2004. Foreign direct investment, imports and innovations in the service industry. Review of Industrial Organization 25(2), 205–227.
- Boehe, D., 2013. Collaborate at home to win abroad: How does access to local network resources influence export behaviour? Journal of Small Business Management 51(2), 167–182.
- Boermans, M.A., Roelfsema, H., 2016. Small firm internationalization, innovation, and growth. International Economics and Economic Policy 13(2), 283–296.
- Bratti, M., Felice, G., 2009. Exporting and Product Innovation at the Firm Level. Mimeo.
- Bratti, M., Felice, G., 2012. Are Exporters More Likely to Introduce Product Innovations? World Economy 35, 1559–1598.
- Brooks, C., 2008. Introductory econometrics for finance. 2nd ed. Cambridge: Cambridge University Press.
- Bustos, P., 2011. Trade liberalization, exports and technology upgrading: evidence on the impact of MERCOSUR on Argentinean firms. American Economic Review 101(1), 304–340.
- Cadogan, J.W., Diamantopoulos, A., Siguaw, J.A., 2002. Export market-oriented activities: Their antecedents and performance consequences. Journal of International Business Studies 33(3), 615–626.
- Calantone, R.J., Cavusgil, S.T., Schmidt, J.B., Shin, G., 2004. Internationalization and the dynamics of product adaptation: an empirical investigation. Journal of Product Innovation Management 21(3), 185–198.
- Cameron, C., Trivedi, P.K., 2005. Microeconometrics: Methods and applications. New York: Cambridge University Press.
- Cassiman, B., & Golovko, E., 2011. Innovation and internationalization through exports. Journal of International Business Studies 42(1), 56–75.
- Cassiman, B., Golovko, E., Martínez-Ros, E., 2010. Innovation, exports and productivity. International Journal of Industrial Organization 28(4), 372–376.
- Cavusgil, S.T., 1998. Knowledge development in international marketing. Journal of International Marketing 6, 103–112.
- Chang, C-L., Chen, S-P., McAleer, M., 2013. Globalization and knowledge spillover: International direct investment, exports and patents. Economic of Innovation and New Technology 22(4), 329–352.
- Chen, Z., Zhang, J., Zheng, W., 2017. Import and innovation: Evidence from Chinese firms. European Economic Review 94, 205–220.
- Cheng, J.L.C., Bolon, D.S., 1993. The management of multinational R&D, a neglected topic in international business research. Journal of International Business Studies 24, 1–18.
- Chiva, R., Ghauri, P., Alegre, J., 2014. Organizational learning, innovation and internationalization: A complex system model. British Journal of Management 25(4), 687–705.
- Cohen, W.M., Levinthal, D.A., 1990. Absorptive capacity: A new perspective on learning and innovation. Administrative Science Quarterly 35(1), 128–152.
- Coe, D., Helpman, E., 1995. International R&D spillovers. European Economic Review 39, 859–87.

- Cooper, R., Kleinschmidt, E., 1985. The impact of export strategy on export sales performance. Journal of International Business Studies, 2, 37–55.
- D'Angelo, A., Majocchi, A., Zucchella, A., Buck, T., 2013. Geographical pathways for SME internationalization: Insights from an Italian sample. International Marketing Review 30(2), 80–105.
- Damijan, J.P., Kostevc, Č., 2010. Learning from Trade through Innovation: Causal Link between Imports, Exports and Innovation in Spanish Microdata. LICOS Discussion Paper, No. 264/2010 (August), Available at SSRN: http://ssrn.com/abstract=1658389 or http://dx.doi.org/1658310.1652139/ssrn.1658389.
- Damijan, J.P., Kostevc, Č., 2015. Learning from trade through innovation. Oxford Bulletin of Economics and Statistics 77(3), 408–436.
- Damijan, J.P., Kostevc, Č., Polanec, S., 2010. From Innovation to Exporting or Vice Versa?' The World Economy 33(3), 374–98.
- Damijan, J.P., Kostevc, Č., Rojec, M., 2017. Exporting status and success in innovation: evidence from CIS micro data for EU countries. The Journal of International Trade & Economic Development 26(5), 585–611.
- Department for Business, Innovation and Skills, (BEIS), 2018a. Longitudinal Small Business Survey, Year One and Two, 2015-2016. [data collection]. 2nd Edition. UK Data Service. SN: 7973, http://doi.org/10.5255/UKDA-SN-7973-2
- Department for Business, Innovation and Skills (BEIS), 2018b. Longitudinal Small Business Survey Year 2 (2016) technical appendix UK Data Service. SN: 7973, http://doc.ukdataservice.ac.uk/doc/7973/mrdoc/pdf/7973_technical_report_2016.pdf
- Dodgson, M., 1993. Organizational learning: A review of some literatures. Organization Studies 14(3), 375–394.
- Erden, Z., Klang, D., Sydler, R., von Krogh, G., 2014. Knowledge flows and firm performance. Journal of Business Research 67(1), 2777–2785.
- Eriksson, K., Chetty, S., 2003. The effect of experience and absorptive capacity on foreign market knowledge. International Business Review 12(6), 673–695.
- Eriksson, K., Johanson, J., Majkgård, A., Sharma, D.D., 2000. Effect of variation on knowledge accumulation in the internationalization process. International Studies of Management & Organization 30(1), 26–44.
- Eriksson, K., Johanson, A., Majkgard, A., Sharma, D., 1997. Experiential knowledge and cost in the internationalization process. Journal of International Business Studies 28(2), 337–360.
- European Commission, 2010. Internationalisation of European SMEs. Brussels: Directorate-General for Enterprise and Industry, European Commission.
- Fassio, C., 2018. Export-led innovation: The role of export destinations. Industrial and Corporate Change 27(1), 149–171.
- Filippetti, A., Frenz, M., Ietto-Gillies, G., 2012. The role of internationalization as a determinant of innovation performance: an analysis of 42 countries. Available at SSRN 2114289.
- Fletcher, M., Harris, S., 2012. Knowledge acquisition for the internationalization of the smaller firm: Content and sources. International Business Review 21(4), 631–647.
- Fletcher, M., 2009. Learning processes in the development of absorptive capacity of internationalising SMEs. In M.V. Jones, P. Dimitratos, M. Fletcher, S. Young (Eds.), Internationalization, Entrepreneurship and the Smaller Firm: Evidence from Around the World, Edward Elgar, Cheltenham, pp.73–90.
- Fletcher, R., 2001. A holistic approach to internationalisation. International Business Review 10(1), 25–49.

- Forsgren, M., 2002. The concept of learning in the Uppsala internationalization process model: a critical review. International Business Review 11(3), 257-277.
- Freel, S., 2000. Barriers to product innovation in small manufacturing firms. International Small Business Journal 18(2), 60–80.
- Fritsch, U., Görg, H., 2015. Outsourcing, Importing and innovation: Evidence from Firm-level data for Emerging Economies. Review of International Economics 23(4), 687–714.
- Ganotakis, P., Love, J.H., 2011. R&D, product innovation, and exporting: Evidence from UK new technology based firms. Oxford Economic Papers: New Series 63, 279–306.
- George, G., Zahra, S.A., Wheatley, K.K., Khan, R., 2001. The effects of alliance portfolio characteristics and absorptive capacity on performance. A study of biotechnology firms. Journal of High Technology Management Research 12(2), 205–266.
- Gerschewski, S., Lew, Y.K., Khan, A., Park, B., 2018. Post-entry performance of international new ventures: The mediating role of learning orientation. International Small Business Journal: Researching Entrepreneurship 36(7), 807–828.
- Gimenez-Fernandez, E.M., Sandulli, F.D., Bogers, M., 2020. Unpacking liabilities of newness and smallness in innovative start-ups: Investigating the differences in innovation performance between new and older small firms. Research Policy 49(10), 104049.
- Goldberg, P.K., Khandelwal, A.K., Pavcnik, N., Topalova, P., 2010. Imported Intermediate Inputs and Domestic Product Growth: Evidence from India. The Quarterly Journal of Economics 125(4), 1727–1767.
- Golovko, E., Valentini, G., 2011. Exploring the complementarity between innovation and export for SMEs growth. Journal of International Business Studies 42(3), 362–380.
- Golovko, E., Valentini, G., 2014. Selective learning-by-exporting: firm size and product versus process innovation. Global Strategy Journal 4, 161–180.
- Gomes, G., Wojahn, R.M., 2017. Organizational learning capability, innovation and performance: study in small and medium-sized enterprises (SMES). Revista de Administração 52, 163–175.
- Grosse, R., Fonseca, A., 2012. Learning through imports in the internationalization process. Journal of International Management 18(4), 366–378.
- Gujarati, D.N., 1995. Basic econometrics (3rd Ed.). Singapore: McGraw-Hill International Editions.
- Gunday, G., Ulusoy, G., Kilic, K., Alpkan, L., 2011. Effects of innovation types on firm performance. International Journal of Production Economics 133(2), 662–676.
- Halilem, N., Amara, N., Landry, R., 2014. Exploring the relationship between innovation and internationalization of small and medium sized enterprises: A nonrecursive structural equation model. Canadian Journal of Administrative Sciences 31, 18–34.
- Hansen, J.A., 1992. Innovation, firm size, and firm age. Small Business Economics 4(1), 37–44.
- Hartman, E., Tower, C., Sebora, T., 1994. Information sources and their relationship to organizational innovation in small business. Journal of Small Business Management 32(1), 10–25.
- Hernández, V., Nieto, M.J., 2016. Inward–outward connections and their impact on firm growth. International Business Review 25, 296–306.
- Higón, D.A., 2011. The impact of ICT on innovation activities: Evidence for UK SMEs. International Small Business Journal 30(6), 684–699.
- Higón, D.A., Driffield, A.N., 2010. Exporting and innovation performance: Analysis of the annual Small Business Survey in the UK. International Small Business Journal 29(1), 4–24.

- Hitt, M.A., Hoskisson, R.E., Kim, H., 1997. International diversification: Effects on innovation and firm performance in product-diversified firms. Academy of Management Journal 40(4), 767–798.
- Huber, G.P., 1991. Organizational learning: The contributing processes and the literatures. Organization Science 2(1), 88-115.
- Hullova, D., Trott, P., Don Simms, C., 2016. Uncovering the reciprocal complementarity between product and process innovation. Research Policy 45, 929–940.
- Iacovone, L., Javorcik, B., 2012. Getting Ready: Preparation for Exporting. Mimeo.
- Idris, B., Saridakis, G., 2018. Local formal interpersonal networks and SMEs internationalisation: Empirical evidence from the UK. International Business Review 27(3), 610–624.
- Jiménez-Jiménez, D., Sanz-Valle, R., 2011. Innovation, organizational learning, and performance. Journal of Business Research 64(4), 408–417.
- Johanson, J., Vahlne, J.E., 2009. The Uppsala internationalization process model revisited: From liability of foreignness to liability of outsidership. Journal of International Business Studies 40(9), 1411–1431.
- Johanson, J., Vahlne, J.E., 1990. The mechanism of internationalisation. International Marketing Review 7(4), 11–24.
- Johanson, J., Vahlne, J.E., 1977. The internationalization process of the firm—a model of knowledge development and increasing foreign market commitments. Journal of International Business Studies 8(1), 23–32.
- Jones, M.V., 2001. First steps in internationalisation: Concepts and evidence from a sample of small high technology firms. Journal of International Management 7(3), 191–210.
- Judge, G.G., Griffiths, W.E., Hill, C., Lütkepohl, H., Lee, T.-C., 1985. The theory and practice of econometrics (2nd ed.). New York: John Wiley.
- Kafouros, M.I., Buckley, P.J., Sharp, J.A., Wang, C., 2008. The role of internationalization in explaining innovation performance. Technovation 28(1/2), 63–74.
- Keller, W., 1998. Are international R&D spillovers trade-related? Analyzing spillovers among randomly matched trade partners. European Economic Review 42(8), 1469–81.
- Kingsley, G., Malecki, E.J., 2004. Networking for competitiveness. Small Business Economics 23, 71–84.
- Kiriyama, N., 2012. Trade and Innovation: Synthesis Report. OECD Trade Policy Papers, No. 135, OECD Publishing, Paris. http://dx.doi.org/10.1787/5k9gwprtbtxn-en
- Korhonen, H., Luostarinen, R., Welch, L., 1996. Internationalization of SMEs: Inward–outward patterns and government policy. Management International Review 36(4), 315–329.
- Kostopoulos, K., Papalexandris, A., Papachroni, M., Ioannou, G., 2011. Absorptive capacity, innovation and financial performance. Journal of Business Research 64(12), 1335–1343.
- Kotabe, M., Murray, J.Y., 1990. Linking product and process innovations and modes of international sourcing in global competition: A case of foreign multinational firms. Journal of International Business Studies 21, 383–408.
- Kotabe, M., Srinivasan, S.S., Aulakh, P.S., 2002. Multinationality and firm performance: The moderating role of R&D and marketing capabilities. Journal of International Business Studies 33(1), 79–97.
- Lecerf, M.A., 2012. Internationalization and innovation: the effects of a strategy mix on the economic performance of French SMEs. International Business Research 5(6), 2–13.
- Lee, J., Lee, J-H., Choi, B., 2014. Effects of Internationalization on Innovation in the Service Industry: Evidence from Korea. Journal of East Asian Economic Integration 18(4), 339–66.

- Levitt, B., March, J.G., 1988. Organizational learning. Annual Review of Sociology 14, 319–340.
- Lewandowska, M.S., Szymura-Tyc, M., Golębiowski, T., 2016. Innovation complementarity, cooperation partners, and new product export: Evidence from Poland. Journal of Business Research 69, 3673–3681.
- Liang, X., Lu, X., Wang, L., 2012. Outward internationalization of private enterprises in China: The effect of competitive advantages and disadvantages compared to home market rivals. Journal of World Business 47, 134–144.
- Li, H., Yi, X., Cui, G., 2017. Emerging market firms' internationalization: how do firms' inward activities affect their outward activities? Strategic Management Journal 38(13), 2704–2725
- Lileeva, A. Trefler, D., 2010. Improved Access to Foreign Markets Raises Plant-level Productivity . . . for Some Plants. The Quarterly Journal of Economics 125(3), 1051–99.
- Liu, Q., Qiu, L.D., 2016. Intermediate input imports and innovations: Evidence from Chinese firms' patent filings. Journal of International Economics 103, 166–183.
- Liu, X., Buck, T., 2007. Innovation Performance and Channels for International Technology Spillovers: Evidence from Chinese High-tech Industries. Research Policy 36(3), 355–66.
- Love, J.H., Ganotakis, P., 2013. Learning by exporting: Lessons from high-technology SMEs. International Business Review 22, 1–17.
- Love, J.H., & Roper, S., 2015. SME innovation, exporting and growth: A review of existing evidence. International Small Business Journal 33(1), 28–48.
- Lu, J.W., Beamish, P.W., 2006. SME Internationalization and performance: Growth vs. profitability. Journal of International Entrepreneurship 27(4), 24–48.
- March, J.G., 1991. Exploration and exploitation in organizational learning. Organization Science 2(1), 71-87.
- Martínez-Ros, E., 2000. Explaining the decisions to carry out product and process innovations: The Spanish case. The Journal of High Technology Management Research 10(2), 223–242.
- Martínez-Ros, E., Labeaga, J.M., 2009. Product and process innovation: Persistence and complementarities. European Management Review 6(1), 64–75.
- Melitz, M., 2003. The impact of trade in intra-industry reallocations and aggregate industry productivity. Econometrica 71(6), 1695–1725.
- Monreal-Pérez, J., Aragón-Sánchez, A., Sánchez-Marín, G., 2012. A longitudinal study of the relationship between export activity and innovation in the Spanish firm: the moderating role of productivity. International Business Review 21(5), 862–877.
- Narayanan, K., Bhat, S., 2009. Technology sourcing and its determinants: a study of Basic Chemical industry in India. Technovation 29(8), 562–573.
- Nguyen, A.N., Pham, N.O., Nguyen, C.D., Nguyen, N.D., 2008. Innovation and exports in Vietnam's SME sector. The European Journal of Development Research 20(2), 262–280.
- Nieto, M.J., Rodríguez, A., 2011. Offshoring of R&D: Looking abroad to improve innovation performance. Journal of International Business Studies 42(3), 345–361.
- Olabisi, M., 2017. The impact of exporting and foreign direct investment on product innovation: Evidence from Chinese manufactures. Contemporary Economic Policy 35(4), 735–750.
- Pangakar, N., 2008. Internationalization and performance of small- and medium-sized enterprises. Journal of World Business 43(4), 475–485.
- Paunov, C., 2011. Imports, Innovation and Employment after Crisis: Evidence from a Developing Country. OECD Science, Technology and Industry Working Papers, 2011/05, OECD Publishing, Paris. http://dx.doi.org/10.1787/5kg0ll7q0fvg-en

- Prashantham, S., 2008. New venture internationalization as strategic renewal. European Management Journal 26(6), 378–387.
- Puthusserry, P., Khan, Z., Knight, G., Miller, K., 2020. How do rapidly internationalizing SMEs learn? Exploring the link between network relationships, learning approaches and post-entry growth of rapidly Internationalizing SMEs from Emerging Markets. Management International Review 60(4), 515–542.
- Quintens, L., Pauwels, P., Matthyssens, P., 2006. Global purchasing: State of the art and research directions. Journal of Purchasing and Supply Management 12(4), 170–181.
- Reichstein, T., Salter, A., 2006. Investigating the sources of process innovation among UK manufacturing. Industrial and Corporate Change 15(4), 653–682.
- Rogers, M., 2004. Networks, firm size and innovation. Small Business Economics 22, 141–153.
- Roper, S., Du, J., Love, J.H., 2008. Modelling the innovation value chain. Research Policy 37(6), 961–977.
- Roper, S., Love, J.H., 2002. Innovation and export performance: Evidence from the UK and German manufacturing plants. Research Policy 31(7), 1087–1102.
- Rosenbaum, P.R., Rubin, D.B., 1983. The Central Role of the Propensity Score in Observational Studies for Causal Effects. Biometrika, 70(1), 41–55.
- Salomon, R. M., Shaver, J.M., 2005. Learning by exporting: New insights from examining firm innovation. Journal of Economics & Management Strategy 14(2), 431–460.
- Saridakis, G., Idris, B., Hansen, J., Dana, L.P., 2019. SMEs' internationalisation: When does innovation matter? Journal of Business Research 96, 250–263.
- Seker, M., 2009. Importing, Exporting, and Innovation in Developing Countries. World Bank, MPRA Paper No. 29904.
- Senge, P., 1990. The Fifth Discipline: The Art and Science of the Learning Organization. New York: Currency Doubleday.
- Serra, F., Pointon, J., Abdou, H., 2012. Factors influencing the propensity to export: A study of UK and Portuguese textile firms. International Business Review 21, 210–224.
- Shahabadi, A., Havaj, S., 2012. The Effect of Technology Spillover through FDI and Import on Innovation. Journal of Rahe Andisheh Economics Research 1(4), 1–20.
- Silva, A., Africano, A.P., Afonso, O., 2010. Do Portuguese manufacturing firms learn by exporting? FEP Working papers 373, Universidade do Portp, Faculdade de Economia do Porto.
- Stinchcombe, A.L., 1965. Social structure and organizations. In: James G. March (eds), *Handbook of Organizations*. Chicago: Rand McNally, pp.142-193.
- Van Beers, C., Zand, F., 2014. R&D cooperation, partner diversity, and innovation performance: An empirical analysis. Journal of Product Innovation Management 31(2), 292–312.
- Vermeulen, F., Barkema, H., 2002. Pace, rhythm, and scope: Process dependence in building a profitable multinational corporation. Strategic Management Journal 23, 637–653.
- Wagner, J., 2002. The causal effects of exports on firm size and labor productivity: first evidence from a matching approach. Economics Letters 77, 287–292.
- Wagner, J., 2007. Exports and productivity: A survey of the evidence from firm-level data. The World Economy 30(1), 60–82.
- Welch, L.S., Benito, G.R.G., Petersen, B., 2007. Foreign operation methods: Theory analysis strategy. Cheltenham: Elgar.
- Welch, L.S., Luostarinen, R.K., 1993. Inward-outward connections in internationalization. Journal of International Marketing 1(1), 44–56.
- Wolff, J.A., Pett, T.L., 2006. Small-Firm Performance: Modelling the Role of Product and Process Improvements. Journal of Small Business Management 44(2), 268–284.

- Westney, D.E., 2020. Rude awakening: threats to the global liberal order. Journal of International Business Studies 51, 1195–1198.
- Yao, Z., Yang, Z., Fisher, G.J., Ma, C., Fang, E., 2013. Knowledge complementarity, knowledge absorption effectiveness, and new product performance: The exploration of international joint ventures in China. International Business Review 22(1), 216–227.
- Yasar, M., Rejesus, R.M., 2005. Exporting status and firm performance: evidence from a matched sample. Economics Letters 88(3), 397–402.
- Zahra, S.A., George, G., 2002. Absorptive capacity: A review, reconceptualization, and extension. Academy of management Review 27(2), 185–203.
- Zahra, S.A., Ucbasaran, D., Newey, L.R., 2009. Social knowledge and SMEs' innovative gains from internationalization. European Management Review 6(1), 81–93.
- Zahra, S.A., Zheng, C., Yu, J., 2018. Learning advantages of newness: A reconceptualization and contingent framework. Journal of International Entrepreneurship 16(1), 12–37.

Appendix A (Tables)

Table A1: Variable definitions used in this study.

Variable	Definition
Overall innovation	Dummy variable = 1 if the firm has introduced product or process innovation.
Innovation types index	Index ranges from 0 to 3: 0 if the firm did not introduce any innovation; 1 if the firm introduced product innovation only; 2 if the firm introduced process innovation only; 3 if the firm introduced both product and process innovation.
Outward internationalisation (i.e., exports)	Whether the firm sells goods and/or services outside the UK (coded 1) or not (coded 0).
Inward internationalisation (i.e., imports)	Whether the firm imports goods and/or services from outside the UK (coded 1) or not (coded 0).
Internationalisation operations	Dummy variable = 1 if the firm is not involved in international operations (i.e. no trade).
	Dummy variable = 1 if the firm exports outside the UK only.
	Dummy variable = 1 if the firm imports from outside the UK only.
	Dummy variable = 1 if the firm both exports and imports from outside the UK.
Firm size	ln(1 + number of employees).
Age of the firm	Broken down into age bands (0–5 years = 1; 6–10 years = 2; 11–20 years = 3; >20 years = 4). Dummy variables are created for each category.
Legal status	Legal status of the business (sole proprietorship = 1, company = 2, partnership = 3). Dummy variables are created for each category.
Sites	Number of sites the business has (1 site = 1; 2 sites = 2; 3 sites = 3; 4–10 sites = 4; 11+ sites = 5). Dummy variables are created for each category.
Turnover	Broken down into turnover bands (1 = less than £82,000; 2 = £82,000–£99,999; 3 = £100,000–£249,000; 4 = £250,000–£499,000; 5 = £500,000–£999,999; 6 = £1 m–£1.99 m; 7 = £2 m–£2.8 m; 8 = £2.81 m–£4.99 m; 9 = £5 m–£9.99 m; 10 = £10 m–£14.99 m; 11 = £15 m–£24.99 m; 12 = £25 m or more). Dummy variables are created for each category.
Business environment: financial obstacle	Dummy variable = 1 if the major obstacle for the business is obtaining finance.
Business environment : intense competition	Dummy variable = 1 if the major obstacle for the business is competition in the local market.
External advice/information	Dummy variable = 1 if the firm sought external advice/information.

R&D tax credit	Dummy variable = 1 if the firm received R&D tax credit in the past three years.
Regions	Location of the business (England = 1, Scotland = 2, Wales = 3, Northern Ireland = 4). Dummy variables are created for each category.
Sectors	SIC 2007 (1-digit) classification. Dummy variables are created for each category.

 Table A2: Descriptive statistics.

	All	Micro firms	Small firms	Medium firms
	SMEs	(<10 employees)	(10-49 employees)	(50-249
Variable				employees)
Overall innovation*	41.52	50.81	30.56	18.63
Innovation combination*				
No innovation (base category)	58.48	64.17	52.21	46.89
Product innovation	19.91	18.54	21.43	22.69
Process innovation	6.40	5.45	7.10	8.99
Product and process innovation	15.21	11.85	19.26	21.43
Outward internationalisation*	21.67	45.96	34.22	19.82
Inward internationalisation**	22.82	44.96	33.39	21.65
Internationalisation operations**				
No international operation (base category)	67.97	73.59	61.73	56.65
Outward internationalisation only	9.21	8.98	9.58	9.51
Inward internationalisation only	10.41	9.58	10.37	13.80
Outward-inward internationalisation	12.41	7.85	18.32	20.03

Notes:

 $[\]begin{array}{l} * \; n_{SMES} = 8171; \; n_{micro} = 4811; \; n_{small} = 2170; \; n_{medium} = 1190. \\ ** \; n_{SMES} = 8139; \; n_{micro} = 4790; \; n_{small} = 2161; \; n_{medium} = 1188. \end{array}$

Table A3: Correlation between the key explanatory variables and the dependent variable (overall innovation) by firm

Variable	All SMEs	Micro firms (<10 employees)	Small firms (10-49 employees)	Medium firms (50-249 employees)
Outward internationalisation†	0.189*	0.166*	0.216*	0.120*
Inward internationalisation‡	0.192*	0.193*	0.171*	0.124*
Internationalisation operations;				
No international operation	-0.221*	-0.213*	-0.233*	-0.122*
Outward internationalisation only	0.077*	0.073*	0.122*	0.006
Inward internationalisation only	0.082*	0.109*	0.051*	0.014
Outward-inward internationalisation	0.169*	0.153*	0.159*	0.133*

Table A4: Correlation between the key explanatory variables and the dependent variable (overall innovation) by firm size.

Variable	All SMEs	Micro firms (<10 employees)	Small firms (10-49 employees)	Medium firms (50-249 employees)
Outward internationalisation†	0.187*	0.163*	0.200*	0.137*
Inward internationalisation‡	0.189*	0.195*	0.152*	0.129*
Internationalisation operations‡				
No international operation	-0.219*	-0.216*	-0.211*	-0.141*
Outward internationalisation only	0.079*	0.074*	0.114*	0.030
Inward internationalisation only	0.082*	0.117*	0.039	0.020
Outward-inward internationalisation	0.164*	0.147*	0.147*	0.135*

Table A5: Propensity score matching results.

Approach	Nearest-neighbour matching			MMWS approach		
Model					Model 1 (T	Table 3)
	Coef.	Std Err.	Coef.	Std Err.	Coef.	Std Err.
Exports	0.160***	0.024			0.119***	0.030
Imports			0.129***	0.026	0.174***	0.022
Exports and imports					0.196***	0.035

Notes: *** p < 0.01.

Appendix B (Figure)

Figure B1: Predictive margins of trade with 95% Cis, n=2,967.

